

## **Author's response to reviews**

**Title:** Study protocol: Evaluation of an online, father-inclusive, universal parenting intervention to reduce child externalising behaviours and improve parenting practices

### **Authors:**

Lucy Tully (lucy.tully@sydney.edu.au)

Patrycja Piotrowska (patrycja.piotrowska@sydney.edu.au)

Daniel Collins (daniel.a.collins@sydney.edu.au)

Kathleen Mairet (kathleen.mairet@gmail.com)

David Hawes (david.hawes@sydney.edu.au)

Eva Kimonis (e.kimonis@unsw.edu.au)

Rhoshel Lenroot (r.lenroot@unsw.edu.au)

Caroline Moul (caroline.moul@sydney.edu.au)

Vicki Anderson (Vicki.Anderson@rch.org.au)

Paul Frick (pfrick@lsu.edu)

Mark Dadds (mark.dadds@sydney.edu.au)

**Version: 2 Date:** 21 May 2017

### **Author's response to reviews:**

Regarding the most recent correspondence from the Editor (16/5/2017), we have now revised the author contributions section of the paper.